Author's response to reviews

Title: The clinical effectiveness of an ultra-brief intervention for common mental health syndromes in primary care: study protocol for a cluster randomised controlled trial

Authors:

Sunny C Collings Prof (sunny.collings@otago.ac.nz)
Fiona Mathieson Snr Lecturer (fiona.mathieson@otago.ac.nz)
Anthony Dowell Prof (tony.dowell@otago.ac.nz)
James Stanley Dr (james.stanley@otago.ac.nz)
Simon Hatcher Dr (shatcher@uottawa.ca)
Felicity A Goodyear-Smith Prof (f.goodyear-smith@auckland.ac.nz)
Brigitte Lane Dr (stress.study@otago.ac.nz)
Amy Munsterman (amyemunsterman@gmail.com)

Version: 5 Date: 11 May 2015

Author's response to reviews: see over
11 May 2015

Dear Editor

Thank you for your email dated 20 April requesting further revisions to the article “The clinical effectiveness of an ultra-brief intervention for common mental health syndromes in primary care: study protocol for a cluster randomised controlled trial”.

I apologise for any confusion caused in responding to this request, and wish to submit the following revisions:

1. Modifications have been made to the manuscript based on the re-review reports. This are detailed separately in the attached document UBI Response to Reviewers SECOND EDITION
2. Affiliations of all authors have been moved to the title page
3. Additional acknowledgements on the title page have been moved to the ‘Acknowledgments and Funding’ section at the end of the article
4. ‘TD’ has been changed to ‘AD’ in the Authors’ Contributions’ section.

Thank you very much for the feedback. Please feel free to contact me if you have any queries.

Best wishes

p.p. Professor Sunny Collings
Dean and Head of Campus
Director, Social Psychiatry & Population Mental Health Research Unit